\PassOptionsToPackage{unicode=true}{hyperref} % options for packages loaded elsewhere
\PassOptionsToPackage{hyphens}{url}
%
\documentclass[]{article}
\usepackage{lmodern}
\usepackage{amssymb,amsmath}
\usepackage{ifxetex,ifluatex}
\usepackage{fixltx2e} % provides \textsubscript
\ifnum 0\ifxetex 1\fi\ifluatex 1\fi=0 % if pdftex
  \usepackage[T1]{fontenc}
  \usepackage[utf8]{inputenc}
  \usepackage{textcomp} % provides euro and other symbols
\else % if luatex or xelatex
  \usepackage{unicode-math}
  \defaultfontfeatures{Ligatures=TeX,Scale=MatchLowercase}
\fi
% use upquote if available, for straight quotes in verbatim environments
\IfFileExists{upquote.sty}{\usepackage{upquote}}{}
% use microtype if available
\IfFileExists{microtype.sty}{%
\usepackage[]{microtype}
\UseMicrotypeSet[protrusion]{basicmath} % disable protrusion for tt fonts
}{}
\IfFileExists{parskip.sty}{%
\usepackage{parskip}
}{% else
\setlength{\parindent}{0pt}
\setlength{\parskip}{6pt plus 2pt minus 1pt}
}
\usepackage{hyperref}
\hypersetup{
            pdftitle={CSci 658-01: Software Language Engineering Spring 2018 Assignments},
            pdfauthor={H. Conrad Cunningham},
            pdfborder={0 0 0},
            breaklinks=true}
\urlstyle{same}  % don't use monospace font for urls
\setlength{\emergencystretch}{3em}  % prevent overfull lines
\providecommand{\tightlist}{%
  \setlength{\itemsep}{0pt}\setlength{\parskip}{0pt}}
\setcounter{secnumdepth}{0}
% Redefines (sub)paragraphs to behave more like sections
\ifx\paragraph\undefined\else
\let\oldparagraph\paragraph
\renewcommand{\paragraph}[1]{\oldparagraph{#1}\mbox{}}
\fi
\ifx\subparagraph\undefined\else
\let\oldsubparagraph\subparagraph
\renewcommand{\subparagraph}[1]{\oldsubparagraph{#1}\mbox{}}
\fi

% set default figure placement to htbp
\makeatletter
\def\fps@figure{htbp}
\makeatother

\usepackage{caption}
\DeclareCaptionLabelFormat{nolabel}{}
\captionsetup{labelformat=nolabel}

\title{CSci 658-01: Software Language Engineering\\
Spring 2018 Assignments}
\author{\textbf{H. Conrad Cunningham}}
\date{\textbf{13 April 2018}}

\begin{document}
\maketitle

Copyright (C) 2018, \href{http://www.cs.olemiss.edu/~hcc}{H. Conrad
Cunningham}

Professor of \href{https://www.cs.olemiss.edu/}{Computer and Information
Science}\\
\href{http://www.olemiss.edu}{University of Mississippi}\\
211 Weir Hall\\
P.O. Box 1848\\
University, MS 38677\\
(662) 915-5358

\hypertarget{general-instructions}{%
\subsection{General Instructions}\label{general-instructions}}

All homework and programming exercises must be prepared in accordance
with the instructions given in the \href{../658syl_sp18.html}{Syllabus}.
Each assignment must be submitted to your instructor by its stated
deadline.

In accordance with expected scholarly and academic standards, if you
reference outside textbooks, reference books, articles, websites, etc.,
or discuss this assignment with individuals inside or outside the class,
you must document these by including appropriate citations or comments
in the header of the primary source file or in another prominent place
in the program documentation.

\hypertarget{descriptions}{%
\subsection{Descriptions}\label{descriptions}}

\begin{enumerate}
\def\labelenumi{\arabic{enumi}.}
\item
  \href{658hw01.html}{Assignment \#1}
\item
  \href{658hw02.html}{Assignment \#2}
\item
  \href{658hw03.html}{Assignment \#3}
\item
  \href{658hw04.html}{Assignment \#4} Final Project
\end{enumerate}

\end{document}
